Author's response to reviews

Title: Developing and implementing an Accreditation system for Health Promoting Schools in Northern India: A cross-sectional study

Authors:

Jarnail S Thakur (jsthakur64@gmail.com)
Deepak Sharma (drdeepakpgimer@gmail.com)
Nidhi Jaswal (nidhi.jaswal@gmail.com)
Bhavneet Bharti (bhavneetb@yahoo.com)
Ashoo Grover (ashoogrover@gmail.com)
Paramjyoti Thind (dr.paramjyoti@gmail.com)

Version: 9 Date: 20 November 2014

Author's response to reviews: see over
To,
The Editor,
BMC Public Health

Subject: - Re-submission of manuscript for publication in BMC Public Health

Sir/Madam,
Please find enclosed our manuscript entitled, ‘Developing and implementing an Accreditation system for Health Promoting Schools in Northern India: A cross-sectional study’ which we would like to resubmit for publication as a research article in journal, ‘BMC Public Health’. We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the manuscript and agree with its submission to **BMC Public Health**.

The comment wise reply to the editor’s comment has been given below:

**Comment 1:** Copyediting: We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz ([www.edanzediting.com/bmc1](http://www.edanzediting.com/bmc1)). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at [http://www.biomedcentral.com/authors/authorfaq/editing](http://www.biomedcentral.com/authors/authorfaq/editing).

**Reply:** We got the copyediting of the manuscript done by the Edanz Editing Services Ltd. as recommended by you. Edited version is being submitted.

**Comment 2:** Please give the full name and affiliation of the ethics committee that approved the study.

**Reply:** The full name and affiliation of the ethics committee is Institutional Ethics Committee (IEC), Postgraduate Institute of Medical Education and Research, Chandigarh

Yours sincerely,

**Corresponding Author,**
Dr. J.S. Thakur, MD
Additional Professor,
School of Public Health,
PGIMER, Chandigarh.
Mobile: 9463602173, Email Id: jsthakur64@gmail.com